Author's response to reviews

Title: Economic burden of diabetes mellitus in the WHO African region

Authors:

Joses M Kirigia (kirigiaj@afro.who.int)
Hama B Sambo (boureima-samboh@afro.who.int)
Luis G Sambo (sambol@afro.who.int)
Saidou P Barry (barrys@afro.who.int)

Version: 6 Date: 30 March 2009

Author's response to reviews:

1. Authors list has been rectified.
2. More information has been provided for web addresses.
3. Full details for references 15 and 24 have been provided
4. The lines in tables have been suppressed using the Microsoft Word facility.
5. The Appendix has been moved to Additional files.
6. Title: the only capitalization are the words 'WHO' and 'African'.
7. Figure has been cropped.